Author's response to reviews

Title: Glomus Tumor in Subcutaneous Layer of Forearm Mimicking Hemangioma: Value of Ultrasonography

Authors:

Dong-Yeong Lee (whatttary@hanmail.net)
Sun-Chul Hwang (hscspine@hanmail.net)
Soon-Taek Jeong (ssurgeon@gsnu.ac.kr)
Dae-Cheol Nam (ortho87@naver.com)
Jin-Sung Park (jsparkler@naver.com)
Jeong-Hee Lee (ihlee7@gnu.ac.kr)
Jae-Boem Na (jbna4216@daum.net)
Dong-Hee Kim (dhkim8311@gnu.ac.kr)

Version: 2 Date: 21 March 2015

Author’s response to reviews:

Glomus tumor is known to commonly affect the hand and it is very rare to manifest in other areas. Since it simulates neuroma, hemangioma and neurofibromatosis, it is difficult to make a differential diagnosis. We conducted marginal resection of solitary mass on forearm which was priorly suspected to be hemangioma or glomus tumor on ultrasound findings, and the tumor was histologically diagnosed to be glomus tumor. We are reporting this case in this article with auxiliary review of literature demonstrating good prognosis of the procedure.